Author’s response to reviews

Title: Exploring the potential for advanced nursing practice role development in Kenya: a qualitative study

Authors:

Linda A East (linda.east@nottingham.ac.uk)
John Arudo (jarudo@gmail.com)
Martha Loefler (Martha.loefler@aku.edu)
Catrin M Evans (Catrin.evans@nottingham.ac.uk)

Version: 3 Date: 8 September 2014

Author's response to reviews:

Dear Sir/Madam

Manuscript number: 2443388821047204 Advanced Nursing Practice in Kenya: exploring the context (now revised to “Exploring the potential for advanced nursing practice role development in Kenya: a qualitative study”)

Please find enclosed a revised article for your consideration for submission in BMC Nursing. The article has been revised following the referees’ comments, for which we are most grateful. A separate file has been uploaded giving a point-by-point response to the referees’ concerns.

The authors’ names and affiliations are as follows:

Dr Linda Anne East, PhD RN
Assistant Professor, The University of Nottingham, United Kingdom
Linda.east@nottingham.ac.uk

Mr John Arudo, B.S.N, M.P.H., MSc
Lecturer, Masinde Muliro University of Science and Technology, Kenya
jarudo@gmail.com

Ms Martha Loefler, B.S.N., MSc
Assistant Professor, Aga Khan University, East Africa
Martha.loefler@aku.edu

Dr Catrin Mai Evans, PhD RN
Associate Professor, The University of Nottingham, United Kingdom
Catrin.evans@nottingham.ac.uk

Correspondence details:
The paper is of interest to the journal as it addresses an issue of contemporary relevance in nursing practice. The research on which the paper is based offers a novel perspective, exploring the context of, and potential for, advanced practice nursing in a Sub-Saharan African setting. We are particularly keen to publish in BMC Nursing as it is an accessible, open-access journal with global reach. We confirm that the manuscript has been submitted solely to the journal and that it is not published, in press, or submitted elsewhere. The names of all the co-authors have been included in the manuscript and that these co-authors all had an active part in the research project.

With thanks for your attention,

Yours sincerely

Dr Linda East